Author's response to reviews

Title: Study protocol Preventing knee injuries in adolescent female football players - design of a cluster randomized controlled trial [NCT00894595]

Authors:

Martin Hägglund (martin.hagglund@liu.se)
Markus Waldén (markus.walden@telia.com)
Isam Atroshi (isam.atroshi@skane.se)

Version: 3 Date: 18 June 2009

Author's response to reviews:

Table 2 has been revised for clarity ("reps" has been removed from the table and specified with a footnote; "sec" was changed to "seconds"). Request; in the original submitted manuscript Table 2 was fitted on one page, while in the proof version of the paper it was on 2 pages. We have not changed this setting in the revised proof, but we would prefer it to be on one page only in the published version.